Author's response to reviews

Title: Emergent approaches to the meta-analysis of multiple heterogeneous complex interventions

Authors:

G.J. Melendez-Torres (gerardo.melendez-torres@spi.ox.ac.uk)
Chris Bonell (c.bonell@ioe.ac.uk)
James Thomas (j.thomas@ioe.ac.uk)

Version: 6 Date: 23 April 2015

Author's response to reviews:

--We have included email addresses of all authors on the title page.
--We have updated authors' affiliations as appropriate.
--We have renamed 'Contributions' as 'Authors' contributions' and 'Statement of competing interest' as 'Competing interests'.
--We removed the section on Funding.
--We have renamed 'Works cited' as References.